Author’s response to reviews

Title: Homocysteine and reactive oxygen species in metabolic syndrome, type 2 diabetes mellitus, and atheroscleropathy: The pleiotropic effects of folate supplementation

Authors:
Dr Melvin R Hayden M.D. (mrh29@usmo.com)
Suresh C Tyagi Ph.D. (s0tyag01@louisville.edu)

Version: 4 Date: 15 Apr 2004

The caps have all been removed from the headings as recommended.

Figure 2 cannot be changed as the wording is embedded in the image and they are not the title. This figure has to remain as it is even though it does not conform to the other power point generated figures.

Thank you for these recommendations and I am sorry that figure 2 has to remain as it was uploaded originally. All the titles to the figures were removed on 30 March 04 and the titles are to be found in the figure legends.

Thank you for your kind help and consideration in making this a better paper.

Sincerely,
M.R. Hayden, M.D.